Contents lists available at ScienceDirect



European research on management and business economics

journal homepage: www.elsevier.es/ermbe



Industry 4.0, servitization, and reshoring: A systematic literature review

Check for updates

Javier Bilbao-Ubillos^{a,*}, Vicente Camino-Beldarrain^b, Gurutze Intxaurburu-Clemente^c, Eva Velasco-Balmaseda^d

^a Department of Applied Economics, Universidad del País Vasco (UPV/EHU), Avda. Lehendakari Agirre, 83, 48015, Bilbao, Spain

^b Department of Applied Economics, Universidad del País Vasco (UPV/EHU), Plaza de Oñati, 1, 20018, Donostia/ San Sebastián, Spain

^c Department of Business Organisation, Universidad del País vasco (UPV/EHU), Plaza de Oñati, 1, 20018, Donostia/ San Sebastián, Spain

^d Department of Business Organisation, Universidad del País vasco (UPV/EHU), C/ Elcano, 21, 48008, Bilbao, Spain

ARTICLE INFO

JEL classification: F02 O32 L60 R11 *Keywords:* Industry 4.0 Servitisation Global value chains Re-shoring Literature review

ABSTRACT

The goal of this literature review is first of all to help define, characterise and contextualise the phenomena that make up Industry 4.0 (I4.0), the servitisation of manufacturing and re-shoring; and secondly to explore the strong interactions between them. We conduct a systemic literature review to identify, synthesise, assess and interpret the findings of past studies to address the research question analysed here. Industry 4.0 technologies (I4.0Ts), and particularly increased digitisation, has made for the configuration of new business models and the servitisation of the economy in the context of a new paradigm of competition. This digital servitisation is conducive to networking and enhances the role of proximity. Together with other concurrent factors (changes in relative costs, agglomeration economies, the skills and expertise in data management required by I4.0Ts) this is favouring reshoring. Technology is always present to a greater or lesser degree as an explanatory factor in re-shoring. The increasing cognitive complexity of technical solutions is enhancing proximity constraints. More frequent, more intense contacts are needed between customers and suppliers in a production set-up that is increasingly customised.

1. Introduction

The progressive implementation of Industry 4.0 technologies (I4.0Ts), particularly increased digitisation, and the servitisation of the economy have made for the configuration of new business models in the context of a new paradigm of competition (Kohtamäki et al., 2022; Weking et al., 2020). Digital servitisation is gradually conditioning global value chains (GVCs) (Kohtamäki et al., 2019; Vendrell-Herrero et al., 2017), and is conducive to networking as it enhances the role of proximity.

Coincidentally, the advent of I4.0Ts changes the cognitive composition of the technology solutions applied in industry and gives rise to new technology needs at production plants (Bilbao-Ubillos et al., 2023). Those needs may often entail territorial constraints in terms of access to the new knowledge now needed to engage in production revealing the advantages of clustering in the digital age when actors look for technological expertise and competences (Jankowska et al., 2021); and servitisation also raises highly specific cognitive issues with consequences especially in terms of territorial constraints. Thus, economic geographic dimension becomes more relevant and relocation decisions -among other firm's strategies- can be expected. In that context, our study thus seeks to determine the significance and scope of the technology element in explaining re-shoring processes.

Baines et al. (2009) define servitisation as "the innovation of an organisation's capabilities and processes to shift from selling products to selling integrated products-services offerings that deliver value in use". According to Neely et al. (2011), servitisation can be summed up in five underlying shifts: 1) from a product-based world to a solution-based world; (2) from products to results; (3) from transactions to relationships; (3) from suppliers to network partners; and (5) from elements to ecosystems.

Digital servitisation describes the convergence of servitisation and digitalization (Gebauer et al., 2021). This convergence permits new ways of value creation (Porter & Heppelmann, 2015). Digital servitisation could thus be considered as the use of digital technologies for transformation processes by which a company moves from a product-centric to a service-centric business model (Kowalkowski et al., 2017).

https://doi.org/10.1016/j.iedeen.2023.100234

Received 25 September 2023; Received in revised form 13 December 2023; Accepted 15 December 2023 Available online 5 January 2024

^{*} Corresponding author. E-mail address: javi.bilbao@ehu.es (J. Bilbao-Ubillos).

^{2444-8834/© 2023} The Author(s). Published by Elsevier España, S.L.U. on behalf of AEDEM. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

The term re-shoring has to do with the decision to relocate manufacturing activities previously moved abroad back to the home country of the parent company (back-shoring) (Kinkel et al., 2021), to a location close to the home country with similar characteristics (near-shoring) (Pedroletti & Ciabuschi, 2023), or to another far away destination (further offshoring) (Elia & Renna, 2023). We use the term "re-shoring" here to describe in a sense equivalent to back-shoring.

There are previous papers that conduct systematic literature reviews (SLR) in the field of digital servitisation (Kohtamäki et al., 2019, 2022; Kolagar et al., 2022; Marcon et al., 2022; Martín-Peña et al., 2018; Paschou et al., 2020), but none of them covers the topic of reshoring. Similarly, previous SLRs in the field of reshoring (Fratocchi et al., 2016, 2019; Wiesmann et al., 2017; Engström et al., 2018; Eriksson et al., 2021) do not address digital servitisation. Thus, we find a marked research gap in the literature: How may I4.0Ts and digital servitisation be affecting business decisions regarding re-shoring? Cosimato & Vona (2021) is the only paper that argues that I4.0Ts applied to industrial processes and engineering applications, together with servitisation, can contribute to the implementation of sustainable reshoring strategies.

Since digital servitisation is a relatively recent, emerging stream of study, an SLR in the fields of I4.0, servitisation and reshoring by pairs is proposed, given that the link between digital servitisation and reshoring has not yet been explored by academic literature. The literature review carried out emphasises the impacts that these phenomena have on territorial dynamics within the framework of the reconfiguration of GVC and answers the following research questions:

RQ1: How has the academic literature on I4.0 and servitisation addressed the impact on territorial dynamics to date?

RQ2: How has the intersection of I4.0 and re-shoring been addressed in academic literature to date?

RQ3: How has the intersection of servitisation and re-shoring been addressed in academic literature to date?

The rest of the paper is organised as follows. Section 2 presents the main concepts sought out in the literature review. Section 3 looks at the review methodology. Section 4 sets out the main points of the pairwise comparison of concepts and summarises the findings. Finally, Section 5 discusses our findings and sets out our conclusions.

2. Conceptual framework for the literature review

2.1. The concept of Industry 4.0

According to Kagermann et al. (2013, p. 14) Industry 4.0 involves the technical integration of Cyber-Physical Systems (CPS) into manufacturing and logistics and the use of the Internet of Things (IoT) and services in industrial processes. In this sense, I4.0 is an umbrella concept for a broad range of technologies and applications to be implemented in relation to different characteristics and performance objectives (Culot et al., 2020). Frank et al. (2019a) divide I4.0Ts according to their main objectives into front-end technologies (Smart Manufacturing, Smart Products, Smart Supply Chain and Smart Working) and base technologies (which include IoT, cloud services, big data and analytics).

Although different terms are used to refer to this phenomenon ("cloud manufacturing", "smart manufacturing", "intelligent manufacturing", etc.), there are commonalities between them (Culot et al., 2020): (1) Key enabling technologies, characterised by increasing digitalisation and connectivity due to physical-digital interface, network, and data-processing technologies and the key role played by the IoT and cloud computing; (2) Distinctive characteristics, which are the technology properties of virtualisation, real-time information sharing, and autonomy, plus greater process integration within and across the company boundaries, interoperability and cybersecurity solutions; (3) Possible outcomes, which mostly relate to higher

productivity and flexibility, making mass customisation possible.

2.2. The concept of servitisation

The term "servitisation" was coined by Vandermerwe & Rada (1988) and means transitioning from the long-established practice of selling products and basic support services to providing advanced combinations of products and services so that greater value is procured for customers by, for example, providing availability. A concept closely linked to servitisation is that of product-service systems (PSS). Tukker (2004) envisages different forms of PSS that include product-oriented services, use-oriented services and result-oriented services.

Neely (2013) sees servitisation as a journey of transformation towards new ways of seeing the relationship between users and suppliers of assets, based more on the provision of services than on the delivery of actual goods, giving rise to cooperation between parties based on long-term contracts.

2.3. The concept of re-shoring

More recently, researchers have detected a number of firms that have reconsidered their previous offshoring decisions and relocated value chain activities, particularly production (Pedroletti & Ciabuschi, 2023). As re-shoring is closely linked to their previous offshoring decisions, when potential benefits of offshoring such as favorable environment, learning opportunities, efficiency and flexibility decrease, firms may start considering the option of bringing back operations to their home country or region in order, for instance, to increase control, offset location problems or even obtain potential new benefits from new policies targeting re-shoring (Pedroletti & Ciabuschi, 2023).

Empirical evidence confirms that these processes have been on the increase in the past 10 years (Dikler, 2021; Raza et al., 2021), and that larger corporations and medium and high-technology industries are most likely to re-shore (Raza et al., 2021). This means that high-end, technology-intensive manufacturing which requires the right infrastructure, skilled workers, intellectual property safeguards and integrated supply chains is increasingly finding that domestic production is the least expensive, most efficient option (Bolter & Robey, 2020). Thus, there is more and more discussion of the vulnerability of GVCs, of the trend for production and sales networks to regionalise more in the context of "slowbalisation" (Raza et al., 2021) and of a regional rebalancing of some GVCs, resulting in a more varied, more scattered production map (De Backer et al., 2016).

3. Methodology: a systematic literature review

We conduct a systemic literature review (SLR) to identify, synthesise, assess and interpret the findings of past studies and address the research question analysed here, i.e. how the academic literature to date on I4.0 and servitisation has addressed the relation between these concepts and reshoring processes. The main value added of this review is the classification of literature, interpretation of findings and identification of gaps.

The field of research addressed by this paper is broad, and covers three main topics: I4.0, digital servitisation and reshoring processes. To conduct a thorough literature review, we decided to split our systemic search into 3 research questions (RQ1, RQ2 & RQ3) and follow a pairwise analysis first to review whether academic literature has addressed each link, and second to analyse what kind of link there is.

We used a combination of strategies to carry out a thorough SLR. On the one hand, we follow the procedures detailed by Dekkers et al. (2022), who set out a multidisciplinary guide to systematic approaches, and the papers of Gatell & Avella (2024), Pucheanu et al. (2022) and Reis et al. (2020), Rejeb et al. (2022), since their SLRs also involve the crossroad between different concepts. On the other hand, to identify additional relevant publications, we also applied snowball and citation search procedures. The SLR comprises three stages: planning and implementing, analysis and classification and reporting of results (Atif et al., 2021; Pucheanu et al., 2022; Reis et al., 2020; Rejeb et al., 2022).

In the first stage, the literature review planning protocol is established and implemented: the research questions are formulated, the key words and terms chosen, the research period set, the search databases selected, and inclusion and exclusion criteria established.

Scopus and the Web of Science (WoS) were selected for the SLR as they are among the most important scientific databases and the documents contained in them are peer-reviewed, which assures their quality (Rejeb et al., 2022). The search was limited to publication dates from 2011, when the term "Industry 4.0" was first coined to 2023. The terms entered in Topic/Keywords for database research are pairwise:

- I. Pair: I4.0 & Servitization (Servitisation);
- II. Pair: I4.0 & Re/Back-shoring/Nearshoring; Smart manufacturing & Re/Back-shoring/Nearshoring; Cloud manufacturing & Re/ Back-shoring/ Nearshoring;
- III. Pair: Servitisation & Re/Back-shoring/Nearshoring; PSS & Re/ Back-shoring/ Nearshoring; Digital Servitisation & Re/Backshoring/Nearshoring; Digital Servitisation & Territorial dynamics; Servitisation & Territorial dynamics; PSS & Territorial dynamics.

As inclusion criteria, "Article", "Book", "Book chapter", and "Review article" written in English were considered, and Meeting, Editorial Material and Conference Proceeding were excluded. For each document, preliminary relevance was determined by title, abstract and keywords. In this stage, duplicated results were fixed. If the paper passed this initial examination, the full reference was obtained for further assessment in the second stage.

In Stage II, preliminary analysis, we reviewed 287 documents' title, abstract and keywords. Inclusion criteria being relevant & nonredundant, 165 papers were selected for full texts reading and extracted information from papers for crosschecking. The inclusion criteria were their links to the main research topic, relevancy and nonredundancy. In each pairwise analysis, we focused on issues related to territorial dynamics, reconfiguration of value chains, supply chains, technology and proximity constraints, to include papers directly or indirectly related to reshoring processes. Documents not close to any of these recurrent topics were excluded as irrelevant: 122 in the preliminary analysis and 34 in the full text reading.

As a complementary snowball and citation review, we identified an initial set of papers and assessed their references and citations. These papers were the following ten documents: Martín-Peña et al., 2018 (cited 218); Paschou et al., 2020 (cited 337); Marcon et al., 2022 (cited 11); Kohtamäki et al., 2019 (cited 719) & 2022 (cited 41); Kolagar et al., 2022 (cited 57), Fratocchi & Di Stefano, 2019 (cited 54) & 2020 (cited 15); Wiesmann et al., 2017 (cited 245); and Cosimato & Vona, 2021 (cited 26). As a result, we included 22 further papers to read, and finally, based on a full reading decided to include 16 of them.

The Stage III of this SLR process, reporting, is summarised in the Fig. 1, and its development and in-depth analysis is presented in the following sections of this article.

4. Results

4.1. RQ1: Industry 4.0 and servitisation

Digital technologies are encouraging manufacturing firms to press ahead with servitisation by speeding up the offering of integrated products and services in the creation of new value and by strengthening relationships with customers in the era of I4.0 (Pirola et al., 2020) and at the same time enabling the implementation of the circular economy paradigm into businesses (Atif et al., 2021; Bressanelli et al., 2020; David et al., 2021; Langley, 2022; Rejeb et al., 2022) and new business models (Agostini & Nosella, 2021; Chen et al., 2021; Kohtamäki et al., 2019; Paiola & Gebauer, 2020; Weking et al., 2020).

According to Paiola & Gebauer (2020), data gathering and analysis is what allows companies to devise a series of servitized value propositions enabled by IoT that are at the base of the redefinitions of the business model. Smart connected products and its generated digital services constitute the fundamental composition of the Smart PSS, which undertake an IT-driven value co-creation manner to fulfil customer needs (Zheng et al., 2019). This results in terms such as "digital servitisation", which refers to the provision of services enabled by ICTs, based on



Fig. 1. Stages and results of SLR. *Source:* Own elaboration.

digital components integrated into physical products (Vendrell-Herrero et al., 2017). Therefore, digital servitisation could become the trigger for a transition from property-based to non-property-based business models, driving advanced revenue models such as pay-per-use, sub-scription or shared use, introducing a completely new value capture mechanism (Paiola & Gebauer, 2020) and establishing reciprocal formulas for sharing risks and rewards between providers and users (Kamp et al., 2017).

Digital servitisation creates business model complexities, such as conflicts between digital and physical service offerings, clashes between new ecosystem partnerships and traditional supply chain relationships, or digital revenue models and product sale models (Chen et al., 2021). The evidence suggests that various categories of competencies allow for different paths of digital servitisation, but that success in digital servitisation requires an appropriate mix of technical, methodological, personal and social skills (Cimini et al., 2021).

In this sense, the advent of digital servitisation has created new challenges in terms of the capabilities needed by companies, with companies' internal capabilities, ecosystem-related and networking capabilities playing an increasing role (Favoretto et al., 2022; Münch et al., 2022; Smania et al., 2022). Thus, digital servitisation transforms the business ecosystem, its supply chains and business relationships (Grandinetti et al., 2020). Therefore, digitisation not only affects the business models of individual companies, but also involves the alignment of the business models of other companies within the ecosystem (Kohtamäki et al., 2019). Ad hoc production networks and the integration of the supply chain are among the main drivers of digital servitisation (Gaiardelli et al., 2021), as the complexity of digital solutions creates a need for customer organisations to consider suppliers as strategic partners working side by side (Sjödin, Kamalaldin, Parida, & Islam, 2023) and intense cooperation between suppliers and (key and selected) customers is a fundamental trait of the most advanced cases of digital servitisation (Grandinetti et al., 2020).

In fact, many digital servitisation initiatives fail due to lack of ecosystem partnerships to ensure value creation (Kolagar et al., 2022). Studies on digital servitisation and business models have emphasized the role played not only by customers, but more broadly, by the actors in the business ecosystem in supporting business model innovations (Agostini & Nosella, 2021), whose governance is considered one of the main challenges. Digitization implies the decoupling of information from devices and technologies, which leads to the dispersion of knowledge creating the need to collaborate, not only with the internal actors of the organization, but also with the actors that are outside the limits of the company (Sklyar et al., 2019). Thus, apart from digitalization capabilities, firms need to develop relational capabilities, such as, providing internal coordination and maintaining external visibility, learning to integrate and coordinate value cocreation activities (Galvani et al., 2022), exploiting current actor competencies through effective knowledge transformation (Kolagar et al., 2022), and design new front-end and back-end roles (Galvani et al., 2022). As a result, issues linked to the culture of transparency and openness, and accountability and trust (Kamp et al., 2017; Kolagar et al., 2022), ethics and security are fundamental in attaining the closest possible convergence between servitisation and I4.0 (Frank et al., 2019b; Kamp et al., 2017).

Finally, firms that implement services benefit from strategic associations with knowledge-intensive service businesses (Bustinza et al., 2017; Paiola et al., 2013). The evidence suggests that the development of smart PSS requires a context of strong technological potential that offers the resources for its development, as well as a support infrastructure and knowledge among stakeholders (Liu et al., 2019; Wang et al., 2022). Moreover, an analysis of the main regional context conditions for 4.0 technological transformation reveals the importance of innovation, education, creativity and entrepreneurship, digital literacy and the presence of a vibrant fabric of industry and local economy (Capello & Lenzi, 2021). This gives rise to the concept of territorial servitisation, which goes beyond organisational limits and brings together the overall results of the various types of mutually dependent association that manufacturing and knowledge-intensive service businesses set up and develop within a specific territory (Lafuente et al., 2017). Interactions between firms and related service businesses at regional level encourage flows of knowledge not just between them, but also to and from various actors in the local value chain. This strengthens both the original fabric of industry and local economic development and resilience, enabling firms also to develop relational and absorption-based capabilities (Gomes et al., 2019).

Table 1 shows the interactions between I4.0 and servitisation detected in our SLR.

4.2. RQ2: Industry 4.0 and re-shoring

Until recently, little attention was paid to the link between I4.0 and re-shoring (Dachs et al., 2019a), but it is now sparking considerable interest in the relevant literature (Ancarani & Di Mauro, 2018; Ancarani et al., 2019; Dachs et al., 2019a; De Backer et al., 2018; Fratocchi & Di Stefano, 2020). The application of I4Ts is bringing about a transformation in manufacturing industry, in terms of both the make-up of processes and the products on offer. Changes are taking place in the cognitive composition of the knowledge bases of the technology solutions used, and in the composition, morphology and geography of value chains, all of which affects re-shoring processes at companies (Bilbao-Ubillos et al., 2021).

First of all, certain territorial dynamics potentially affecting relocation decisions should be considered: Stentoft et al. (2021) indicate that the qualifications required to handle Industry 4.0 technologies may decrease the willingness of firms offshore and increase the likelihood of re-shoring. Kinkel et al. (2023) also highlight the importance of digital

Table 1

Interactions between I4.0 and servitisation in the SLR.

Digitalization as facilitators of servitisation	Porter & Heppelmann, 2015; Grubic, 2018; Ardolino et al., 2018; Kohtamäki et al., 2019; Zheng et al., 2019; Coreynen et al., 2020; Grandinetti et al., 2020; Bilbao-Ubillos et al., 2021; Vendrell-Herrero et al., 2021; Niu & Qin, 2021; Paiola et al., 2022
Cloud computing, big data or Internet of Things as facilitators of servitisation	Bressanelli et al., 2018; Morelli et al., 2022; Niu & Qin, 2021; Paiola & Gebauer, 2020; Paiola et al., 2021; Paschou et al., 2020; Peng, 2020; Reis et al., 2020, 2020; Wen & Zhou, 2016.
Role of other I4.0Ts in servitisation processes	Neely, 2013; Frank et al., 2019b; Zheng et al., 2019; Liu et al., 2019; Pirola et al., 2020; Gaiardelli et al., 2021; Cimini et al., 2021
Paths for servitisation & benefits	Bustinza et al., 2017; Coreynen et al., 2020; Galvani et al., 2022; Gomes et al., 2019; Kamp et al., 2017; Kohtamäki et al., 2020; Lafuente et al., 2017; Legault et al., 2019; Paiola & Gebauer, 2020; Paiola et al., 2013.
Digital Servitization & Business Ecosystem	Kohtamäki et al., 2019; Sklyar et al., 2019; Grandinetti et al., 2020; Sjödin, Kamalaldin, Parida, & Islam, 2023; Galvani et al., 2022; Münch et al., 2022; Smania et al., 2022
Digital Servitization & Business Models	Bressanelli et al., 2018; Kohtamäki et al., 2019; Weking et al., 2020; Martín-Peña et al., 2020; Agostini & Nosella, 2021; Chen et al., 2021; Paiola & Gebauer, 2020; Paiola et al., 2021, 2022; Sivula et al., 2022
Digital servitization & capabilities/ competences	Cimini et al., 2021; Coreynen et al., 2020; Khan et al., 2022; Münch et al., 2022; Smania et al., 2022
Digital servitization & circular economy	Atif et al., 2021; Bressanelli et al., 2018, 2020; David et al., 2021; Kolagar et al., 2022; Langley, 2022; Rejeb et al., 2022.

Source: Own elaboration.

skills in the operation of new technologies associated with artificial intelligence (AI) and point out that they act as drivers fo back-shoring decisions.

In line with the operational complexity involved in handling Industry 4.0 technologies, Lampón et al. (2017) highlight the importance of technology-related demands in the design of relocation strategies as a result of the resources and capacities that host territories must offer if they are to handle new technologies efficiently. Barbieri et al. (2022) highlight that the development of policies to drive Industry 4.0 provides an opportunity to generate location advantages that can attract home-country firms located abroad.

But aside from these studies, which have in common the facts that technology is a determinant factor in location and that the development of Industry 4.0 is conducive to back-shoring. Bailey & De Propris (2014) highlight other aspects: They focus on the case of the UK automotive industry and state that a combination of a more competitive exchange rate (despite the very recent appreciation of sterling), increased transport costs, rising wages in key areas of China, and a greater awareness of supply chain resilience have all contributed to a perceived change in some business 'fundamentals'. The potential for some supply chain re-shoring also links in with the trend of the 'servitisation' of manufacturing, including the auto sector, and shift to a hybrid model where manufacturing and services are increasingly intertwined (Bailey & De Propris, 2014, 60).

But in our opinion, technology and know-how are determinant factors in re-shoring decisions for the following reasons: First, there is a greater combinatorial and technological complexity in support knowledge due to the incorporation of the new knowledge fields needed by the process of digitalisation of manufacturing production. This incorporation of new knowledge fields also increases the cognitive content of tasks, given that new technologies must be incorporated in a specific fashion with specific technical results required in the context of the new solution drawn up. Any effort to adapt technically will always require a process of investigation which is specific to each product/process. Making such specific adaptations calls for sufficient technology potential to carry out the necessary cognitive synthesis.

Secondly, the new knowledge entailed by digital technologies is science-based and features a high degree of novelty (latest generation), so it can be expected to have a large tacit component. This is liable to involve proximity constraints for efficient transfer. Moreover, the great scope of digital technologies for generating new applications in all directions means that they can be expected to develop rapidly, offering new, more efficient, more comprehensive practical possibilities and calling continually for the adaptation of technology (Bilbao-Ubillos et al., 2021).

Thirdly, changes in the cognitive content of technical solutions affect both the organisation and composition of value chains (with participants potentially joining and leaving), changing the relationships established between participants. This not only conditions the shape of value chains but also shifts them towards more complex forms of governance that entail more intensive proximity constraints (Gereffi & Lee, 2012). Thus, in re-shoring decisions there is a need to assess the risk of losing technical knowledge and access to knowledge (both within the firm and in the local network), as this could act as a barrier to re-shoring (Boffelli & Johansson, 2020). Some experiences show a positive relationship between the success of the re-shoring strategy through I4.0 and the preparation of the local ecosystem, so that the company is able to absorb the tacit knowledge of its local system (Pegoraro et al., 2022).

There is agreement in economic literature that the effects of digitalisation on employment are marked by two basic tendencies: in one jobs are reduced as a result of process automation, and in the other there is an increase in the qualifications required of human resources to work with the new products and processes (Autor, 2015; Bonekamp & Sure, 2015).

The greater cognitive complexity of the technical solutions in manufacturing poses a problem of adaptation to new production requirements. There is a need for personnel trained in new qualifications and the skills needed to assimilate new knowledge and adapt it to new technical solutions so as to operate efficiently. This in turn calls for a cognitive rethinking at businesses in which priority must be given to personnel training (Kamp & Gamboa, 2021; Kohtamäki et al., 2019). As the core role of digital technologies grows, this may necessitate the implementation of a digital culture at organisations.

Therefore, the setting where production is located must have a supply of labour with the required skill sets (Eurofound, 2019b; Strange & Zucchella, 2018), because skill shortages are an issue to be considered in re-shoring (Boffelli & Johansson, 2020; Lampón & Gonzalez-Benito, 2020; Moradlou et al., 2022). Thus, countries performing well in human capital, education, innovation, network readiness and digitalisation will be well placed to take advantage of re-shoring decisions (Bonekamp & Sure, 2015).

In the last 2 decades authors insist on the changing patterns of production relocation (Fan & Liu, 2021). Some developed countries have proposed plans for "re-industrialization" and "manufacturing recovery" to restore their own economies and adopted different levels of protectionist measures (Stentoft et al., 2016).

The cases covered in economic literature confirm that technology is a key explanatory factor in re-shoring. Stentoft & Stegmann (2014) conclude that the need for greater proximity between R&D and production to take advantage of technological opportunities is one of the conditioning factors for re-shoring decisions. Thus, re-shoring is justified by the increasing importance of innovation and quality, which can be boosted through the location of R&D and production centres closer to the end customer, and by the need to safeguard intellectual property rights.

Kinkel (2014) and Dachs et al. (2019a) highlight quality and the flexibility to respond rapidly to customised demand as key factors for re-shoring, but transport and coordination costs, staff qualification problems and proximity to R&D activities are also important (Kinkel, 2014, 64; Dachs et al., 2019b). Kamp & Gibaja (2021) do not find any clear impact of I4.0Ts on backshoring operations in their study, but Ancarani et al. (2019), Cosimato & Vona (2021) and Barbieri et al. (2018) observe a link between re-shoring and I4.0Ts. Ketokivi et al. (2017) observe that the technologically highly complex activities that involve highly specific knowledge, with low levels of codification, are more likely to re-shore. Lampón & Rivo-López (2021) find that in highly technology-intensive industries an innovation-oriented strategy drives re-shoring strategies, while in low-technology sectors a cost-oriented strategy prevails.

Moser (2013) states that the loss of operational flexibility, the reduction in innovation potential and difficulties in finding qualified personnel are among the basic reasons that drive re-shoring processes. I4.0 thus encourages re-shoring whenever bringing the value chain back together is important for developing products, creating prototypes, etc. In such cases, joint location of production and development may become a source of value creation, and new manufacturing technologies can help reduce production costs (Ancarani & Di Mauro, 2018).

Once it is realised that there are generally links between I4.0 and reshoring, it must also be understood that the various technologies involved in I4.0 may have different effects on re-shoring (Ancarani & Di Mauro, 2018; Fratocchi & Di Stefano, 2020; Raza et al., 2021). In industries with higher labour costs, locating to developed countries has only recently become economically viable because of the increasing degree of process automation (Stentoft et al., 2017). However, Raza et al. (2021) state that automation can also be conducive to offshoring, as it helps to strengthen the productivity of production processes in emerging economies. And Ancarani & Di Mauro (2018) conclude that robotics is not a necessary ingredient for re-shoring.

Of the various technologies involved in I4.0, robotics is best suited to customised manufacturing, benefiting industries where market demand and consumer preferences change rapidly. In such industries, robots enable firms to bring new products to the market far more quickly than with offshoring, which often involves distant countries where suppliers do not always produce to the correct specifications, resulting in quality problems and long lead times (De Backer et al., 2018).

Big data analysis and digitalisation increase the ability to coordinate operations remotely, which may make re-locating to low-cost countries an attractive prospect (Ancarani & Di Mauro, 2018), but additive manufacturing technologies have a positive impact on decisions to re-shore, pushing value chains to become more local and move closer to end-users (Fratocchi, 2017; Moradlou & Tate, 2018; Raza et al., 2021). This is especially true of re-shoring decisions made on quality grounds (Ancarani & Di Mauro, 2018). Specifically, Moradlou & Tate (2018) identify 6 areas where additive manufacturing may potentially impact the supply chains of companies, making them more receptive to re-shoring: shorter lead times, better response in terms of product and market changes, lower transport costs, fewer communication errors with suppliers, more customisation options and lower product inventories in stock.

As can be seen (Table 2), academic interest in the link between reshoring and I4.0 has grown over time, though for the moment there is limited empirical evidence (Dachs et al., 2019a; Raza et al., 2021).

4.3. RQ3: servitisation and re-shoring

The symbiosis between digitalisation and servitisation has major cognitive consequences for business operations (Peng, 2020; Ruiz de la Torre & Sánchez, 2022). The high levels of innovation and the scientific basis of supporting knowledge (Pucheanu et al., 2022; Zheng et al., 2019) mean that the tacit component of the knowledge used acquires a core role, which leads to major proximity constraints in transfer processes. Despite the benefits of better communication, some types of knowledge are so tacit in nature that only spatial proximity (e.g. co-location) can enable them to be absorbed by actors throughout the value chain (Baldoni et al., 2022). In fact, in the case of SMEs, servitised firms are more likely than non-servitised firms to engage in innovation activities, to use absorptive capabilities in the appropriation of external knowledge and to collaborate with partners in the innovation process (Punstein & Glückler, 2021).

At the same time, servitisation processes encourage the reorganising of production operations based on platforms that can support the customer and supplier integration needed to achieve efficient collaboration on technology between all stakeholders (Grandinetti et al., 2020; Niu & Qin, 2021; Zheng et al., 2019). Such collaboration is needed to adapt to faster technical dynamics and be open to changes in business models, so as to take advantage of the opportunities offered by changing circumstances (Barbieri et al., 2020; Bolter & Robey, 2020; Dachs et al., 2019b; Di Mauro et al., 2018; Eurofound, 2019a; Paiola et al., 2022; Weking et al., 2020). This conditions the shape of value chains and brings about a shift towards more complex forms of governance in which proximity constraints play a greater role (Gereffi & Lee, 2012).

The business models used will call for ever-higher levels of skill and qualifications among customers and suppliers too, and for powerful telematic infrastructures (both public and private) to support new operational requirements (Liu et al., 2019; Wang et al., 2022). This in turn calls for physical proximity between stakeholders as a key part of assuring rapid response and the opportunity to acquire critical knowledge of customers' operations and needs (Strange & Zucchella, 2018). It also requires customers and suppliers to be located in a setting where the technology resources needed to take part in an integrated, advanced process are available (Sjödin, Kamalaldin, Parida, & Islam, 2023).

The literature that analyses reshoring processes on the basis of case studies confirms the link between digital servitisation and reshoring. These findings show the core role of the technical and organisation requirements associated with digital servitisation in explaining reshoring. The need to make good use of knowledge transfers to production plants (Ashby, 2016; Fjellström et al., 2023; Sayem et al., 2019), the need to be close to customers and markets (Fjellström et al., 2019), access to human resources with the necessary qualifications and skills (Boffelli et al., 2021; Lampon & Gonzalez-Benito, 2020; Moradlou et al., 2022; Moser, 2013), the efficient reorganisation of supply chains (Park & Hong, 2017; Robinson & Hsieh, 2016; Van Hoek & Dobrzykowski, 2021) and the need for R&D and production operations to be close to each other (Bals et al., 2016; Pearce, 2014) are also relevant factors. All these circumstances share the need to locate production plans in a setting where they can adapt their technologies to new production requirements (Lund & Steen, 2020; Pegoraro et al., 2022). Reshoring is an effective way of doing this.

Finally, secure data exchange (Kagermann & Wahlster 2022) and cybersecurity are an issues raised by digital servitization; in some cases, reshoring is a better option than offshoring in terms of control, cost and risk (Engström et al., 2018; Moretto et al., 2020; Park & Hong, 2017; Sayem et al., 2019). Another factor is the customer's perception of value, which still has a pivotal role in back-shoring decisions (Moradlou et al., 2022), as does ethnocentrism, i.e. consumers' tendency to exhibit a positive predisposition toward products made in their home country (Grappi et al., 2015).

Table 3 shows the main general interactions between servitisation and reshoring detected in the SLR.

4.4. Synthesis of findings

The SLR conducted focuses on pairwise comparison of the three thematic terms. It enables us to explore how I4.0Ts and servitisation may be affecting business decisions in regard to re-shoring (Fig. 2).

As the SLR shows, the scenario entailed by I4.0Ts and servitisation strategies intensifies the need for greater proximity and closer collaboration between the actors involved in generating new, innovative solutions, as the cognitive complexity of the knowledge bases of the technology solutions will make companies more dependent on external networks and will generate proximity constraints. Servitised designs are set to increase the intensity and frequency of customer/supplier relations greatly, and in the framework of the sharing of highly-novel, scientifically-based knowledge they entail strong proximity

Table 2

nteractions between I4.0 and Re-shoring in the SLR.					
Cause & effect relationship	Empirical studies on interaction	Case Studies confirming the relation	Case Studies not confirming the relation		
Arlbjørn & Mikkelsen, 2014; Saki, 2016; Ancarani & Di Mauro, 2018; Ancarani et al., 2019; Barbieri et al., 2018; De Backer et al., 2018; Eurofound, 2019a & 2019b; Fratocchi & Di Stefano, 2020; Moradlou & Tate, 2018; Strange & Zucchella, 2018; Stentoft & Rajkumar, 2020; Boffelli & Johansson, 2020; Moser, 2021; Ancarani et al., 2021; Cosimato & Vona, 2021.	Laplume et al., 2016; Stentoft et al., 2017; Lampón et al., 2017; Ancarani & Di Mauro, 2018; Moradlou & Tate, 2018; Dachs et al., 2019a; Fratocchi & Di Stefano, 2020; Stentoff et al., 2021; Raza et al., 2021; Unterberger & Müller, 2021; Barbieri et al., 2022; Kinkel et al., 2023.	Eurofound, 2019a; Fratocchi, 2017; Gylling et al., 2015; Thomas Industrial Survey, 2020; Ancarani et al., 2021; Bolter & Robey, 2020; Lampon & Gonzalez-Benito, 2020. Raza et al., 2021; Dikler, 2021; Moradlou et al., 2022.	Kamp & Gibaja, 2021.		

Source: Own elaboration.

Table 3

Interactions between Servitisation and Re-shoring in the SLR.

Cause & effect relationship	Empirical studies on interaction
Effciency of R&D: concentration of resources; use of R&D transferred to plants. Proximity between R&D and production	Ashby, 2016; Bals et al., 2016; Fjellström et al., 2023; Khan et al., 2022; Pearce II, 2014; Sayem et al., 2019. Hills, 2017; Schmeisser, 2013; Urzelai & Puig, 2017; Barbieri et al., 2020; Bolter & Robey, 2020; Dachs et al., 2019b; Di Mauro et al., 2018; Paiola et al., 2022; Weking et al., 2020.
Qualifications among human resources & supply of technological resources in settingcontext Reorganisation of the value chain	Moser, 2013; Ancarani et al., 2015; Bals et al., 2016; Baraldi et al., 2018; Di Mauro et al., 2018; Liu et al., 2019; Lund & Steen, 2020; Sansone et al., 2020; Boffelli et al., 2021; Lampon & Gonzalez-Benito, 2020; Martinez-Mora & Merino, 2021; Moradlou et al., 2022; Opazo-Basaez et al., 2022; Pegoraro et al., 2022; Wang et al., 2022; Bals et al., 2016; Grandinetti et al., 2020; Gylling et al., 2015; Niu & Qin, 2021; Park & Hong, 2017; Pearce II, 2014; Robinson & Hsieh, 2016; Sjödin, Kamalaldin, Parida, & Islam, 2023; Van Hoek & Dobrzikowski, 2021; Zhong et al., 2019.
Business model.	Córcoles et al., 2021; Grubic, 2018; Vendrell-Herrero et al., 2017.
Better customer service; flexibility; delivery times	Baines et al., 2009; Kinkel, 2014; Martinez-Mora & Merino, 2014; Rabetino et al., 2017; Moradlou et al., 2017; Fjellström et al., 2019; Johansson et al., 2019; Martinez-Mora & Merino, 2021; Wisemen et al., 2017

Source: Own elaboration.

constraints, which become a very powerful conditioning factor for location.

On the other hand, I4.0Ts lead to digital servitisation processes in which the technological capabilities of companies and the territories where they are located acquire great relevance. They require settings with high-level technology resources and human skills, and entail a need to increase collaboration between ecosystem actors, all of which may be expected to have a major impact on re-shoring decisions.

5. Discussion of findings and conclusions

5.1. Servitisation and setting

Servitisation increases the complexity of the production system via the incorporation of I4.0Ts into corporate operations in the form of a more extensive, deeper integration. This also affects firms' relations with technology-centred institutions and with customers and suppliers. Efficient servitisation entails close links with customers and suppliers that go beyond new technical assistance. Customers and suppliers will report needs, propose solutions, etc. that will have far-reaching consequences in terms of location, because there will be circumstances that require proximity for cooperation to be efficient, and clients and suppliers will need to located in a setting where they have the potential to provide themselves with the resources to take part in an integrated, advanced process (Sjödin, Kamalaldin, Parida, & Islam, 2023).

The technology setting will play a core role in the supply of human and technological resources, and in providing a fabric of production that meets the requirements for the business model set up to be efficient. The resources offered by the setting will mark the potential for advancement in developing smart manufacturing capabilities (Khan et al., 2022; Vaillant et al., 2021). A weak setting forces firms to opt for specialised solutions with a more limited scope, which may be of interest in the short term but offer less of a future (Legault et al., 2019). The choice of a location for industrial operations is an increasingly selective process (Taleb et al., 2022). In short, location becomes increasingly important as servitisation becomes more sophisticated and calls for a more complex, more efficient organisational model.

5.2. Servitisation and proximity constraints

As emerging technologies with a high degree of novelty, the development of servitisation is characterised by the prominent role played by basic knowledge, and thus by a substantial tacit component in information transfer processes. The rate of change in the technical solutions arising from the huge capacity of I4.0Ts for generating new applications thus necessitates frequent, rapid changes. This in turn necessitates continual recourse to settings capable of handling the necessary adaptations. The tacit component and the frequency of interactions with external factors strengthen the role of proximity in the factors involved in the dynamics of technology (Baldoni et al., 2022). Moreover, the wealth of recombinations offered by I4.0Ts makes for a scenario with high levels of uncertainty in terms of goals and paths. This highlights the importance of locating in settings with a high potential for the provision of I4.0Ts that can cater for as-yet-undefined future cognitive needs. Proximity ensures a scenario in which interactions multiply to give rise to unexpected new solutions (De la Torre & Sanchez, 2022; Foray & Goenaga, 2013).

The extension of servitisation increases the need for integration with the other agents needed for it to operate. Technology-related problems are not limited to the adaptation and development of new technical solutions at corporate laboratories but extend to the make-up of links with all the public and private sector organisations associated with the operation of companies, especially customers and suppliers. Links in which the cognitive content has a high tacit component and there is continual interaction strengthen proximity constraints between agents and, equally importantly, call for a localised fabric of customers and suppliers in a setting where there is a high level of I4.0Ts on offer. This can be thought of as a highly powerful territory effect that justifies the emphasis placed by Mattes (2012) on geographical proximity as a facilitator of other forms of proximity (social, organisational and institutional) which are important for the efficient operation of activities that require a high level of integration between participants.

5.3. Conclusions

5.3.1. Theoretical contribution

I4.0Ts, and particularly increased digitisation, has made for the configuration of new business models and the servitisation of the economy in the context of a new paradigm of competition. This digital servitisation is conducive to networking and enhances the role of proximity. Together with other concurrent factors, this is favouring reshoring. The process drivers directly concerned with technology are the following:

- Changes in relative costs: the automation of processes entailed by I4.0 is reducing the amount of labour required per unit of product and thus the significance of labour costs as a factor for offshoring.
- Agglomeration economies entailed by digital servitisation. On the one hand there is the greater density of firms and the greater complementarity between them that can be found in more developed countries (which are generally the sources of capital), which is significant for PSS. On the other hand there is the need for technology infrastructure in the host country (technology centres, engineering firms), which is increased by the greater combinatorial complexity of the supply of firms.
- The advantages of proximity to end markets entailed by digital servitisation: the ability to respond quickly and flexibly to market changes, manufacture customised products, shorten development periods and reduce waste. Offshoring took production plants away from developed countries, so that the potential that now emerges from customisation arising from digital servitisation was lost. The

- 14.0Ts demand technological resources (technology services, technology advice, training, qualified personnel, ec)
- Changes in the cognitive composition of the knowledge bases of technology solutions:
 - ✓ Greater combinatorial complexity in support knowledge (technological potential to carry out cognitive synthesis)
 - ✓ Knowledge entailed by digital technologies is scientifically-base and features high degree of novelty → Tacit component; Proximity Constraints
 - ✓ Change in the Value Chain: more complex forms of governance → Intensive proximity constraints

Smart PSS requires a context of strong technological potential that offers the resources for its development, a support infrastructure and knowledge among stakeholders In digital servitisation suppliers, customers and other partners become part of an ecosystem network around cyber-physical systems Decoupling of information from devices and technologies, which leads to the dispersion of knowledge: collaboration needed among internal an external actors in digital servitisation	 Technological and organizational requirements of digital servitisation Tacit component of the knowledge originates proximity restrictions Higher level of interaction end customer/service provider and supplier integration Settings with high levels of skills and qualifications and technology resources Secure data exchange and cybersecurity issues 	RESHORING
	Smart PSS requires a context of strong technological potential that offers the resources for its development, a support infrastructure and knowledge among stakeholders In digital servitisation suppliers, customers and other partners become part of an ecosystem network around cyber-physical systems Decoupling of information from devices and technologies, which leads to the dispersion of knowledge: collaboration needed among internal an external actors in digital servitisation	 Smart PSS requires a context of strong technological potential that offers the resources for its development, a support infrastructure and knowledge among stakeholders In digital servitisation suppliers, customers and other partners become part of an ecosystem network around cyber-physical systems Decoupling of information from devices and technologies, which leads to the dispersion of knowledge: collaboration needed among internal an external actors in digital servitisation Technological and organizational requirements of digital servitisation Tacit component of the knowledge originates proximity restrictions Higher level of interaction end customer/service provider and supplier integration Settings with high levels of skills and qualifications and technology resources Secure data exchange and cybersecurity issues

Fig. 2. Interactions among Industry 4.0, servitisation and reshoring. *Source:* Own elaboration.

demand for better customer service leads to a need for greater proximity because the content of that service is cognitively more complex and the market for more evolved (servitised) production processes lies precisely in higher income countries (the source countries of the investment made abroad for cost reasons). This is conducive to reshoring. In activities where emerging markets are also significant it is possible that not all production may be reshored, but at least part of it will be, so as to provide a fast response to customers in developed countries.

- The skills and expertise in data management required by I4.0Ts: wider availability in the developed countries that tend to be sources of FDI.

In turn, these factors conducive to proximity benefit from the need to address uncertainties in exchange rates, interest rates, trading policies (the tendency towards protectionism), in the economic consequences of potential conflicts or pandemics (supply chain disruption) and in undesirable inventory levels. These uncertainties make locating production in other settings a less attractive prospect.

Factors that offset this push towards re-shoring include sunk costs and the need of component suppliers to be close to the plants of their customers when they work on a just-in-time basis, or when the end product of firms is aimed at distant markets.

Implications for research and implications in practice:

This study focuses research and corporate decision-making on the role of proximity in the new technology setting that stems from digital servitisation, and therefore it provides managers with a detailed overview of the main factors for making re-shoring decisions.

The most significant implication for public authorities is concerned with the development of technological support in their reference territories to cater for training of personnel in digital skills and fostering and supporting technical assistance bodies in matters of digital technologies. In other words, the challenge is to design a context of support for production operations based on digital technologies, so as to prevent firms from leaving and encourage firms to set up in their areas (reshoring).

5.3.2. Limitations and future research

Our work is not without limitations. First, SLR applies a wellstructured, rigorous method. However, some studies may potentially have been excluded due to research decisions such as database and keyword selection, publication type and language. In any case, we believe that the publications identified are representative of the current literature. Second, I4.0 is a general concept for a wide range of technologies and applications, and each of those technologies could have a different impact on servitisation and reshoring. Therefore, future research could focus on the study of these relationships. More specifically, how do the different I4.0Ts affect digital servitisation and reshoring processes? Do all the technologies that fall under I4.0 work in the same direction?

Our study offers indications of the potential links between digital servitisation and re-shoring in general, without establishing nuances depending on the type of company in question. Therefore, another important line of research could be to study the link between digital

References

servitisation and re-shoring depending on the sector of activity of the company or its position in the value chain. Finally, recent studies show the role of I4.0 in the implementation of the circular economy in the servitisation business model (Atif et al., 2021). In the context of these business models, to what extent will the drive towards the circular economy boost re-shoring processes?

Funding

The authors acknowledge research funding from Basque Government (GIC21/113 BILBAO UBILLOS, JAVIER).

CRediT authorship contribution statement

Javier Bilbao-Ubillos: Conceptualization, Project administration, Supervision, Writing – review & editing. Vicente Camino-Beldarrain: Methodology, Formal analysis, Writing – original draft, Writing – review & editing. Gurutze Intxaurburu-Clemente: Writing – original draft, Writing – review & editing. Eva Velasco-Balmaseda: Investigation, Data curation.

Declaration of Competing Interest

The authors declare that they have no conflict of interest.

Appendix A. Selected documents in the literature review

Agostini & Nosella (2021), Ancarani & Di Mauro (2018), Ancarani et al. (2015), Ancarani et al. (2019), Ancarani et al. (2021), Ardolino et al. (2018), Arlbjørn & Mikkelsen (2014), Ashby (2016), Atif et al. (2021), Baines et al. (2009), Bals et al. (2016), Baraldi et al. (2018), Barbieri et al. (2018, 2020, 2022), Bilbao-Ubillos et al. (2021), Boffelli & Johansson (2020), Boffelli et al. (2021), Bolter & Robey (2020), Bressanelli et al. (2020), Bressanelli et al. (2018), Bustinza et al. (2017), Chen et al. (2021), Cimini et al. (2021), Córcoles et al. (2021), Coreynen et al. (2020), Cosimato & Vona (2021), Dachs et al. (2019a, 2019b), David et al. (2021), De Backer et al. (2018), Di Mauro et al. (2018), Dikler (2021), Eurofound (2019a, 2019b), Fjellström et al. (2019, 2023), Frank et al. (2019b), Fratocchi (2017), Fratocchi & Di Stefano (2020), Gaiardelli et al. (2021), Galvani et al. (2022), Gomes et al. (2019), Grandinetti et al. (2020), Grubic (2018), Gylling et al. (2015), Hills (2017), Johansson et al. (2019), Kamp & Gibaja (2021), Kamp et al. (2017), Khan et al. (2022), Kinkel (2014), Kinkel et al. (2023), Kohtamäki et al. (2019), Kohtamäki et al. (2020), Kolagar et al. (2022), Lafuente et al. (2017), Lampon & Gonzalez-Benito (2020), Lampón et al. (2017), Langley (2022), Laplume et al. (2016), Legault et al. (2019), Liu et al. (2019), Lund & Steen (2020), Martín-Peña et al. (2020), Martinez-Mora & Merino (2014, 2021), Moradlou & Tate (2018), Moradlou et al. (2017), Moradlou et al. (2022), Morelli et al. (2022), Moser (2021, 2013), Münch et al. (2022), Neely (2013), Niu & Qin (2021), Opazo-Basaez et al. (2022), Paiola & Gebauer (2020), Paiola et al. (2013, 2022), Paiola et al. (2021), Park & Hong (2017), Paschou et al. (2020), Pearce (2014), Pegoraro et al. (2022), Peng (2020), Pirola et al. (2020), Porter & Heppelmann (2015), Rabetino et al. (2017), Raza et al. (2021), Reis et al. (2020), Rejeb et al. (2022), Robinson & Hsieh (2016), Saki (2016), Sansone et al. (2020), Sayem et al. (2019), Schmeisser (2013), Sivula et al. (2022), Sjödin et al. (2023), Sklyar et al. (2019), Smania et al. (2022), Stentoft et al. (2017, 2021), Stentoft & Rajkumar (2020), Strange & Zucchella (2018), Thomas Industrial Survey (2020), Unterberger & Müller (2021), Urzelai & Puig (2017), Van Hoek & Dobrzykowski (2021), Vendrell-Herrero et al. (2017), Vendrell-Herrero et al. (2021), Wang et al. (2022), Weking et al. (2020), Wen & Zhou (2016), Wiesmann et al. (2017), Zheng et al. (2019).

Agostini, L., & Nosella, A. (2021). Industry 4.0 and business models: A bibliometric literature review. Business Process Management Journal, 27(5), 1633–1655.

- Ancarani, A., & Di Mauro, C. (2018). Re-shoring and Industry 4.0: How often do they go together? IEEE Engineering Management Review, 46(2), 87–96.
- Ancarani, A., Di Mauro, C., & Mascali, F. (2019). Back-shoring strategy and the adoption of Industry 4.0: Evidence from Europe. *Journal of World Business*, 54(4), 360–371.
- Ancarani, A., Di Mauro, C., Fratocchi, L., Orzes, G., & Sartor, M. (2015). Prior to reshoring: A duration analysis of foreign manufacturing ventures. *International Journal of Production Economics*, 169, 141–155.
- Ancarani, A., Di Mauro, C., Virtanen, Y., & You, W. (2021). From China to the west: Why manufacturing locates in developed countries. *International Journal of Production Research*, 59(5), 1435–1449.
- Ardolino, M., Rapaccini, M., Saccani, N., Gaiardelli, P., Crespi, G., & Ruggeri, C. (2018). The role of digital technologies for the service transformation of industrial companies. *International Journal of Production Research*, 56(6), 2116–2132.
- Arlbjørn, J. S., & Mikkelsen, O. S. (2014). Backshoring manufacturing: Notes on an important but under-researched theme. *Journal of Purchasing and Supply Management*, 20(1), 60–62.
- Ashby, A. (2016). From global to local: Reshoring for sustainability. Operations Management Research, 9, 75–88.
- Atif, S., Ahmed, S., Wasim, M., Zeb, B., Pervez, Z., & Quinn, L. (2021). Towards a conceptual development of Industry 4.0, servitisation, and circular economy: A systematic literature review. *Sustainability*, 13(11), 6501.
- Autor, D. H. (2015). Why are there still so many jobs? The history and future of workplace automation. *The Journal of Economic Perspectives*, 29(3), 3–30.
- Bailey, D., & De Propris, L. (2014). Reshoring: Opportunities and limits for manufacturing in the UK—The case of the auto sector. *Revue d'Economie Industrielle*, 145(1), 45–61.
- Baines, T., Lightfoot, H., Peppard, J., Johnson, M., Tiwari, A., Shehab, E., & Swink, M. (2009). Towards an operations strategy for product-centric servitization. *International Journal of Operations and Production Management*, 29(5), 494–519.
- Baldoni, E., De Propris, L., & Storai, D. (2022). Territorial servitisation in Italian industrial districts: The role of technological and professional KIBS. Competitiveness Review: An International Business Journal, 32(5), 743–759.
- Bals, L., Kirchoff, J., & Foerst, K. (2016). Exploring the re-shoring and insourcing decision making process: Toward an agenda for future research. *Operations Management Research*, 9, 102–116.
- Baraldi, E., Ciabuschi, F., Lindahl, O., & Fratocchi, L. (2018). A network perspective on the reshoring process: The relevance of the home-and the host-country contexts. *Industrial Marketing Management*, 70, 156–166.
- Barbieri, P., Boffelli, A., Elia, S., Fratocchi, L., Kalchschmidt, M., & Samson, D. (2020). What can we learn about re-shoring after COVID-19? *Operations Management Research*, 13(3), 131–136.
- Barbieri, P., Boffelli, A., Elia, S., Fratocchi, L., & Kalchschmidt, M. (2022). How does Industry 4.0 affect international exposure? The interplay between firm innovation and home-country policies in post-offshoring relocation decisions. *International Business Review*, 31(4), Article 101992.
- Barbieri, P., Ciabuschi, F., Fratocchi, L., & Vignoli, M. (2018). What do we know about manufacturing reshoring? *Journal of Global Operations and Strategic Sourcing*, 11(1), 79–122.
- Bilbao-Ubillos, J., Camino-Beldarrain, V., & Intxaurburu-Clemente, G. (2021). Industry 4.0, proximity constraints and new challenges for industrial policy. *European Planning Studies*, 29(2), 329–345.
- Bilbao-Ubillos, J., Camino-Beldarrain, V., Intxaurburu-Clemente, G., & Velasco-Balmaseda, E. (2023). Industry 4.0 and potential for reshoring: A typology of technology profiles of manufacturing firms. *Computers in Industry*, 148, Article 103904.
- Boffelli, A., & Johansson, M. (2020). What do we want to know about re-shoring? Towards a comprehensive framework based on a meta-synthesis. *Operations Management Research*, 13(1), 53–69.
- Boffelli, A., Fratocchi, L., Kalchschmidt, M., Silva, S., & Lima da Costa, C. (2021). Doing the right thing or doing things right: What is better for a successful manufacturing reshoring? Operations Management Research, 14, 1–16.
- Bolter, K., & Robey, J. (2020). Strategic re-shoring: A literature review. Prepared for The Fund for our Economic Future (FFEF). Available at <u>Https://research.upjohn.org/re ports/253</u>.
- Bonekamp, L., & Sure, M. (2015). Consequences of Industry 4.0 on human labour and work organization. Journal of Business Media Psychology, 6, 33–40.
- Bressanelli, G., Adrodegari, F., Perona, M., & Saccani, N. (2018). Exploring how usagefocused business models enable circular economy through digital technologies. *Sustainability*, 10(3), 639.
- Bressanelli, G., Saccani, N., Perona, M., & Baccanelli, I. (2020). Towards circular economy in the household appliance industry: An overview of cases. *Resources*, 9 (11), 128.
- Bustinza, O. F., Gomes, E., Vendrell-Herrero, F., & Baines, T. (2017). Product–service innovation and performance: The role of collaborative partnerships and R&D intensity. *R&D Management*, 49(1), 33–45.
- Córcoles, D., Díaz Mora, C., & Gandoy, R. (2021). Servitización y digitalización de las manufacturas españolas. *Economía Industrial*, 422, 25–35.
- Capello, R., & Lenzi, C. (2021). Industry 4.0 and servitisation: Regional patterns of 4.0 technological transformations in Europe. *Technological Forecasting and Social Change*, 173, Article 121164.
- Chen, Y., Visnjic, I., Parida, V., & Zhang, Z. (2021). On the road to digital servitization—The (dis) continuous interplay between business model and digital

J. Bilbao-Ubillos et al.

technology. International Journal of Operations and Production Management, 41(5), 694-722.

- Cimini, C., Adrodegari, F., Paschou, T., Rondini, A., & Pezzotta, G. (2021). Digital servitization and competence development: A case-study research. CIRP Journal of Manufacturing Science and Technology, 32, 447–460.
- Coreynen, W., Matthyssens, P., Vanderstraeten, J., & van Witteloostuijn, A. (2020). Unravelling the internal and external drivers of digital servitization: A dynamic capabilities and contingency perspective on firm strategy. *Industrial Marketing Management*, 89, 265–277.
- Cosimato, S., & Vona, R. (2021). Digital innovation for the sustainability of reshoring strategies: A literature review. Sustainability, 13(14), 7601.
- Culot, G., Nassimbeni, G., Orzes, G., & Sartor, M. (2020). Behind the definition of Industry 4.0: Analysis and open questions. *International Journal of Production Economics*, 226, Article 107617.
- Dachs, B., Kinkel, S., & Jäger, A. (2019a). Bringing it all back home? Back-shoring of manufacturing activities and the adoption of Industry 4.0 technologies. *Journal of World Business*, 54(6), Article 101017.
- Dachs, B., Kinkel, S., Jäger, A., & Palčič, I. (2019b). Back-shoring of production activities in European manufacturing. *Journal of Purchasing and Supply Management*, 25(3), Article 100531.
- David, G. M., Druta, R. M., Birgovan, A. L., Bacali, L., & Lungu, F. (2021). Industry 4.0 and the circular economy: A systematic review of the literature. Acta Technica Napocensis-Series: Applied Mathematics, Mechanics, and Engineering, 64(4s), 651–656.
- De Backer, K., DeStefano, T., Menon, C., & Suh, R.J. (2018). "Industrial robotics and the global organisation of production." OECD Science, Technology and Industry Working Papers, 2018/03. Paris. OECD Publishing.
- De Backer, K., Menon, C., Desnoyers-James, I., & Moussiegt, L. (2016). Re-shoring: Myth or reality?. In OECD science, technology and industry policy papers, 27 Paris: OECD Publishing.
- De la Torre, A. R., & Sanchez, D. (2022). Evolution of Servitization: New business model opportunities. International Journal of Production Management and Engineering, 10(1), 77–90.
- Dekkers, R., Carey, L., & Langhorne, P. (2022). Making literature reviews work: A multidisciplinary guide to systematic approaches. Springer.
- Di Mauro, C., Fratocchi, L., Orzes, G., & Sartor, M. (2018). Offshoring and back-shoring: A multiple case study analysis. *Journal of Purchasing and Supply management*, 24, 108–124.
- Dikler, J. (2021). Re-shoring: An overview, recent trends, and predictions for the future. KIEP Research Paper, World Economy Brief, 21–35. Available at SSRN: https://ssrn. com/abstract=3916557 or https://doi.org/10.2139/ssrn.3916557.
- Elia, S., & Renna, G. (2023). Offshoring vs further offshoring: A multi-level analysis on the firm-, country- and industry-specific drivers of business model reconfiguration. In S. T. Marinova, & M. A.. Marinov (Eds.), *Reconfiguration of business models and ecosystems. decoupling and resilience* (pp. 79–165). N. York: Routledge.
- Engström, G., Sollander, K., Hilletofth, P., & Eriksson, D. (2018). Reshoring drivers and barriers in the Swedish manufacturing industry. *Journal of Global Operations and Strategic Sourcing*, 11(2), 174–201.
- Eriksson, D., Hilletofth, P., Tate, W. L., & Göthager, M. (2021). Critical manufacturing prerequisites for successful reshoring. *Operations and Supply Chain Management*, 14 (2), 249–260.
- Eurofound. (2019a). *Re-shoring in Europe: Overview 2015–2018*. Luxembourg: Publications Office of the European Union.
- Eurofound. (2019b). *The future of manufacturing in Europe*. Luxembourg: Publications Office of the European Union.
- Fan, X-m., & Liu, H-g. (2021). Global supply chain shifting: A macro sense of production relocation based on multi-regional input-output table. *Economic Modelling*, 94, 672–680.
- Favoretto, C., Mendes, G. H., Oliveira, M. G., Cauchick-Miguel, P. A., & Coreynen, W. (2022). From servitization to digital servitization: How digitalization transforms companies' transition towards services. *Industrial Marketing Management*, 102, 104–121.
- Fjellström, D., Hilletofth, P., Fang, T., Kumar, V., Kumar, A., & Tan, K. (2023). Manufacturing relocation ambiguity model: A prerequisite for knowledge management. *British Journal of Management*, 34(3), 1100–1116.
- Fjellström, D., Fang, T., & Chimenson, D. (2019). Explaining reshoring in the context of Asian competitiveness: Evidence from a Swedish firm. *Journal of Asia Business Studies*, 13(2), 277–293.
- Foray, D., & Goenaga, X. (2013). "The goals of smart specialization" S3 policy brief series no. 01/2013 – May, S3 platform. Seville: Institute for Prospective Technological Studies-Joint Research Centre (IPTS-JRC).
- Frank, A. G., Dalenogare, L. S., & Ayala, N. S. (2019a). Industry 4.0 technologies: Implementation patterns in manufacturing companies. *International Journal of Production Economics*, 210, 15–26.
- Frank, A. G., Mendes, G. H., Ayala, N. F., & Ghezzi, A. (2019b). Servitization and Industry 4.0 convergence in the digital transformation of product firms: A business model innovation perspective. *Technological Forecasting and Social Change*, 141, 341–351.
- Fratocchi, L., & Di Stefano, C. (2019). Does sustainability matter for reshoring strategies? A literature review. Journal of Global Operations and Strategic Sourcing, 12(3), 449–476.
- Fratocchi, L., & Di Stefano, C. (2020). Do Industry 4.0 technologies matter when companies backshore manufacturing activities? An explorative study comparing Europe and the US. In M. Bettiol, E. Di Maria, & S. Micelli (Eds.), *Knowledge* management and Industry 4.0. knowledge management and organizational learning (pp. 53–83). Berlin: Springer.
- Fratocchi, L., Ancarani, A., Barbieri, P., Di Mauro, C., Nassimbeni, G., Sartor, M., ... Zanoni, A. (2016). Motivations of manufacturing reshoring: an interpretative

framework. International Journal of Physical Distribution & Logistics Management, 46 (2), 98–127.

- Fratocchi, L. (2017). Is 3D printing an enabling technology for manufacturing reshoring? In A. Vecchi (Ed.), Reshoring of manufacturing: Drivers, opportunities, and challenges (pp. 99–124). Cham: Springer.
- Gaiardelli, P., Pezzotta, G., Rondini, A., Romero, D., Jarrahi, F., Bertoni, M., et al. (2021). Product-service systems evolution in the era of Industry 4.0. *Service Business*, 15(1), 177–207.
- Galvani, S., Carloni, E., Bocconcelli, R., & Pagano, A. (2022). From after-sales to advanced services: A network analysis on the impacts of digital servitization evolution. *Sustainability*, *14*(14), 8308.
- Gatell, I. S., & Avella, L. (2024). Impact of Industry 4.0 and circular economy on lean culture and leadership: Assessing digital green lean as a new concept. European Research on Management and Business Economics, 30(1), Article 100232.
- Gebauer, H., Paiola, M., Saccani, N., & Rapaccini, M. (2021). Digital servitization: Crossing the perspectives of digitization and servitization. *Industrial Marketing Management*, 93, 382–388.
- Gereffi, G., & Lee, J. (2012). Why the world suddenly cares about global supply chains. Journal of Supply Chain Management, 48(3), 24–32.
- Gomes, E., Bustinza, O. F., Tarba, S., Khan, Z., & Ahammad, M. (2019). Antecedents and implications of territorial servitization. *Regional Studies*, 53(3), 410–423.

Grandinetti, R., Ciasullo, M. V., Paiola, M., & Schiavone, F. (2020). Fourth industrial revolution, digital servitization and relationship quality in Italian B2B manufacturing firms. An exploratory study. *The TQM Journal*, 32(4), 647–671.

Grappi, S., Romani, S., & Bagozzi, R. P. (2015). Consumer stakeholder responses to reshoring strategies. Journal of the Academy of Marketing Science, 43(4), 453–471.

Grubic, T. (2018). Remote monitoring technology and servitization: Exploring the relationship. *Computers in Industry*, 100, 148–158.

- Gylling, M., Heikkilä, J., Jussila, K., & Saarinen, M. (2015). Making decisions on offshore outsourcing and backshoring: A case study in the bicycle industry. *International Production Economics*, 162, 92–100.
- Hills, K. (2017). "Meet The Maker: The SockMine Make Moves to Reshore Their Sock Manufacturing." Available at: <u>Https://makeitbritish.co.uk/uk-factories/sock-manufacturing/</u>.
- Thomas Industrial Survey. (2020, April). COVID-19's impact on North American manufacturing. Thomasnet. Available at: https://www.thomasnet.com/insights/m anufacturer-response-to-covid-19-disruptions-increased-interest-in-automatio n-reshoring/.

Jankowska, B., Di Maria, E., & Cygler, J. (2021). Do clusters matter for foreign subsidiaries in the Era of industry 4.0? The case of the aviation valley in Poland. *European Research on Management and Business Economics*, 27(2), Article 100150.

Johansson, M., Olhagera, J., Heikkiläb, J., & Stentoft, J. (2019). Offshoring versus backshoring: Empirically derived bundles of relocation drivers, and their relationship with benefits. *Journal of Purchasing and Supply Management*, 25(3), Article 100509.

- Kagermann, H., & Wahlster, W. (2022). Ten Years of Industrie 4.0. Science, 4(3), 26. Kagermann, H., Helbig, J., Hellinger, A., & Wahlster, W. (2013). Recommendations for implementing the strategic initiative Industrie 4.0: Securing the future of German manufacturing industry; final report of the Industrie 4.0 working group. Forschungsunion.
- Kamp, B., & Gamboa, J. (2021). Industry 4.0 technologies, skills and training and their influence on the servitization of industrial firms. In Servitization: "A pathway towards a resilient productive and sustainable future". Proceedings of the spring servitization conference 2021.
- Kamp, B., & Gibaja, J. J. (2021). Adoption of digital technologies and backshoring decisions: Is there a link? Operations Management Research, 14(3), 380–402.
- Kamp, B., Ochoa, A., & Diaz, J. (2017). Smart servitization within the context of industrial user-supplier relationships: Contingencies according to a machine tool manufacturer. *International Journal on Interactive Design and Manufacturing*, 11(3), 651–663.
- Ketokivi, M., Turkulainen, V., Seppälä, T., Rouvinen, P., & Ali-Yrkkö, J. (2017). Why locate manufacturing in a high-cost country? A case study of 35 production location decisions. *Journal of Operations Management*, 49–51, 20–30.

Khan, M. A., Stoll, O., West, S., & Wuest, T. (2022). Equipment upgrade service provision in the context of servitization: Drivers, capabilities, and resources. *Production Planning and Control*, 1–19. https://doi.org/10.1080/09537287.2022.2063199

- Kinkel, S. (2014). Future and impact of back-shoring-Some conclusions from 15 years of research on German practices. *Journal of Purchasing and Supply management, 20*, 63–65.
- Kinkel, S., Capestro, M., Di Maria, E., & Bettiol, M. (2023). Artificial intelligence and relocation of production activities: An empirical cross-national study. *International Journal of Production Economics*, 261, Article 108890.

Kinkel,S., Capestro. M. & Di Maria, E. (2021). Artificial Intelligence and Backshoring Strategies: A German-Italian Comparison, in Bryan Christiansen and Tihana Škrinjarić: Handbook of Research on Applied AI for International Business and Marketing Applications. Hershey (Pennsylvania). IGI Global. (227-255).

Kohtamäki, M., Parida, V., Oghazi, .. P., Gebauer, H., & Baines, T. (2019). Digital servitization business models in ecosystems: A theory of the firm. *Journal of Business Research*, 104, 380–392.

- Kohtamäki, M., Rabetino, R., Parida, V., Sjödin, D., & Henneberg, S. (2022). Managing digital servitization toward smart solutions: Framing the connections between technologies, business models, and ecosystems. *Industrial Marketing Management*, 105, 253–267.
- Kohtamäki, M., Parida, V., Patel, P. C., & Gebauer, H. (2020). The relationship between digitalization and servitization: The role of servitization in capturing the financial

J. Bilbao-Ubillos et al.

potential of digitalization. Technological Forecasting and Social Change, 151, Article 119804.

Kolagar, M., Parida, V., & Sjödin, D. (2022). Ecosystem transformation for digital servitization: A systematic review, integrative framework, and future research agenda. Journal of Business Research, 146, 176-200.

Kowalkowski, C., Gebauer, H., & Oliva, R. (2017). Service growth in product firms: Past, present, and future. Industrial Marketing Management, 60, 82-88.

Lafuente, E., Vaillant, Y., & Vendrell-Herrero, F. (2017). Territorial servitization: Exploring the virtuous circle connecting knowledge-intensive services and new manufacturing businesses. International Journal of Production Economics, 192, 19-28. Lampón, J. F., & Rivo-López, E. (2021). The effect of the industry technology intensity on the drivers of manufacturing backshoring. Journal of Manufacturing Technology

Management, 33(1), 1–21. Lampón, J. F., Cabanelas, P., & Carballo-Cruz, F. (2017). A model for international production relocation: Multinationals' operational flexibility and requirements at production plant level. Journal of Business Research, 77, 95-101.

Lampon, J. F., & Gonzalez-Benito, J. (2020). Backshoring and improved key manufacturing resources in firms' home location. International Journal of Production Research, 58(20), 6268-6282.

Langley, D. J. (2022). Digital product-service systems: The role of data in the transition to servitization business models. Sustainability, 14(3), 1303.

Laplume, A. O., Petersen, B., & Pearce, J. M. (2016). Global value chains from a 3D printing perspective. Journal of International Business Studies, 47(5), 595-609.

Legault, P., de Santa-Eulalia, L. A., Mosconi, E., Bordeleau, F.È., Francoeur, C., Cadieux, N., Anholon, R., & Rekik, Z. (2019). Servitization trend in the machinetools market: Comparing value from turnkey and specialized IoT-based analytics solutions using TOPSIS. Procedia Manufacturing, 31, 390–397.

Liu, B., Zhang, Y., Zhang, G., & Zheng, P. (2019). Edge-cloud orchestration driven industrial smart product-service systems solution design based on CPS and IioT. Advanced Engineering Informatics, 42, Article 100984.

Lund, H., & Steen, M. (2020). Make at home or abroad? Manufacturing reshoring through a GPN lens: A Norwegian case study. Geoforum, 113, 154-164.

Münch, C., Marx, E., Benz, L., Hartmann, E., & Matzner, M. (2022). Capabilities of digital servitization: Evidence from the socio-technical systems theory. Technological Forecasting and Social Change, 176, Article 121361.

Marcon, É., Marcon, A., Ayala, N. F., Frank, A. G., Story, V., Burton, J., Raddats, C., & Zolkiewski, J. (2022). Capabilities supporting digital servitization: A multi-actor perspective. Industrial Marketing Management, 103, 97–116.

Martín-Peña, M. L., Díaz-Garrido, E., & Sánchez-López, J. M. (2018). The digitalization and servitization of manufacturing: A review on digital business model. Strategic Change, 27(2), 91-99.

Martín-Peña, M.-L., Sánchez-López, J.-M., & Díaz-Garrido, E. (2020). Servitization and digitalization in manufacturing: the influence on firm performance. Journal of Business & Industrial Marketing, 35(3), 564-574.

Martinez-Mora, C., & Merino, F. (2014). Offshoring in the Spanish footwear industry: A return journey. Journal of Purchasing and Supply Management, 20, 225-237.

Martinez-Mora, C., & Merino, F. (2021). Extending the offshoring literature To explain backshoring: An application to the Spanish footwear industry. Growth and change, 53 (3), 1230 - 1250.

Mattes, J. (2012). Dimensions of proximity and knowledge bases: Innovation between spatial and non-spatial factors. Regional Studies, 46(8), 1085-1099.

Moradlou, H., & Tate, W. (2018). Re-shoring and additive manufacturing. World Review of IntermodalTransportation Research, 7(3), 241–263.

Moradlou, H., Backhouse, C. J., & Ranganathan, R. (2017). Responsiveness, the primary reason behind re-shoring manufacturing activities to the UK: An Indian industry perspective. International Journal of Physical Distribution & Logistics Management, 47 (2-3), 222-236.

Moradlou, H., Fratocchi, L., Skipworth, H.& ., & Ghadge, A. (2022). Post-Brexit backshoring strategies: What UK manufacturing companies could learn from the past? Production Planning and Control, 33(14), 1319–1336.

Morelli, G., Magazzino, C., Gurrieri, A. R., Pozzi, C., & Mele, M. (2022). Designing smart energy systems in an Industry 4.0 paradigm towards sustainable environment. Sustainability, 14(6), 3315.

Moretto, A., Patrucco, A. S., & Harland, C. M. (2020). The dynamics of reshoring decisions and the role of purchasing. International Journal of Production Research, 58 (19) 5929–5944

Moser, H. (2013). Manufacturing: Re-shoring tools to retain companies and attract foreign direct investment. Economic Development Journal, 12(1), 5-11.

Moser, H. (2021). COVID-19 Drives Industry 4.0-and Reshoring. Society of Manufacturing Engineer Blog (entry August 30, 2021). Available at: https://www. sme.org/technologies/articles/2021/august/covid-19-drives-industry-4.0-and-re shoring/

Neely, A. (2013).: "What is Servitization?" Andy Neely. Blog, 30-November-2013. Available at: Http://andyneely.blogspot.com.es/2013/11/what-is-servitization. html.

Neely, A., Benedettini, O., & Visnjic, I. (2011). The servitization of manufacturing: Further evidence. Paper presented at the 18th European operations management association conference. 1, Cambridge.

Niu, X., & Qin, S. (2021). Integrating crowd-/service-sourcing into digital twin for advanced manufacturing service innovation. Advanced Engineering Informatics, 50, Article 101422.

Opazo-Basaez, M., Vendrell-Herrero, F., & Bustinza, O. (2022). Digital service innovation: A paradigm shift in technological innovation. Journal of Service Management, 33(1), 97–120.

Paiola, M., & Gebauer, H. (2020). Internet of things technologies, digital servitization and business model innovation in BtoB manufacturing firms. Industrial Marketing Management, 89, 245-264.

Paiola, M., Agostini, L., Grandinetti, R., & Nosella, A. (2022). The process of business model innovation driven by IoT: Exploring the case of incumbent SMEs. Industrial Marketing Management, 103, 30-46.

Paiola, M., Saccani, N., Perona, M., & Gebauer, H. (2013). Moving from products to solutions: Strategic approaches for developing capabilities. European Management Journal, 31(4), 390-409.

Paiola, M., Schiavone, F., Khvatova, T., & Grandinetti, R. (2021). Prior knowledge, Industry 4.0 and digital servitization. An inductive framework. Technological Forecasting and Social Change, 171, Article 120963.

Park, Y., & Hong, P. (2017). Reshoring strategy: Case illustrations of japanese manufacturing firms. In A. Vecchi (Ed.), Reshoring of manufacturing: Drivers, opportunities, and challenges (pp. 143-161). Cham: Springer.

Paschou, T., Rapaccini, M., Adrodegari, F., & Saccani, N. (2020). Digital servitization in manufacturing: A systematic literature review and research agenda. Industrial Marketing Management, 89, 278–292.

Pearce, J., II (2014). Why domestic outsourcing is leading America's reemergence in global manufacturing. Business Horizons, 57, 27-36.

Pedroletti, C., & Ciabuschi, F. (2023). Reshoring: A review and research agenda. Journal of Business Research, 164, Article 114005.

Pegoraro, D., De Propris, L., & Chidlow, A. (2022). Regional factors enabling manufacturing reshoring strategies: A case study perspective. Journal of International Business Policy, 5(1), 112-133.

Peng, S. (2020). A new trade regime for the servitization of manufacturing: Rethinking the goods-services dichotomy. Journal of World Trade, 54(5), 699-726.

Pirola, F., Boucher, X., Wiesner, S., & Pezzotta, G. (2020). Digital technologies in product-service systems: A literature review and a research agenda. Computers in Industry, 123, Article 103301.

Porter, M. E., & Heppelmann, J. E. (2015). How smart, connected products are transforming companies. Harvard Business Review, 93(10), 96-114.

Pucheanu, F., Colesca, S., Pacesila, M., & Burcea, S. (2022). Industry 4.0, sustainability and innovation at the crossroad: A review analisys. Management Research and Practice, 14(3), 47-69.

Punstein, A. M., & Glückler, J. (2021). How small and medium sized firms walk the path to hybridity. *Sustainability*, *13*(5), 2511. Rabetino, R., Kohtamäki, M., & Gebauer, H. (2017). Strategy map of servitization.

International Journal of Production Economics, 192, 144–156.

Raza, W., Grumiller, J., Grohs, H., Essletzbichler, J., & Pintar, N. (2021). Post COVID-19 value chain: Options for re-shoring production back to Europe in a globalised economy. Brussels: European Union-Directorate General for External Policies of the Union.

Reis, A., Júnior, E., Baumann, B., & Halbe, M. (2020). Prospects for using gamification in Industry 4.0. *Production*, *30*(6), Article e20190094.

Rejeb, A., Suhaiza, Z., Rejeb, K., Seuring, S., & Treiblmaier, H. (2022). The Internet of Things and the circular economy: A systematic literature review and research agenda. Journal of Cleaner Production, 350, Article 131439.

Robinson, P., & Hsieh, L. (2016). Reshoring: A strategic renewal of luxury clothing supply chains. Operations Management Research, 9, 89–101.

Buiz de la Torre A & Sanchez D (2022) Evolution of Servitization: New business model opportunities. International Journal of Production Management and Engineering, 10(1), 77–90.

Saki, Z. (2016). Disruptive innovations in manufacturing–an alternative for re-shoring strategy. Journal of Textile and Apparel. Technology and Management, 10(2), 1-7.

Sansone, C., Hilletofth, P., & Eriksson, D. (2020). Critical operations capabilities for competitive manufacturing in a high-cost environment: a multiple case study. Operations and Supply Chain Management: An International Journal, 13(1), 94-107.

Sayem, A., Feldmann, A., & Ortega-Mier, M. (2019). Investigating the influence of network-manufacturing capabilities to the phenomenon of reshoring: An insight from three case studies. BRQ Business Research Quarterly, 22, 68-82.

Schmeisser, B. (2013). A systematic review of literature on offshoring of value chain activities. Journal of Internatinal Management, 19(4), 390-406.

Sivula, A., Shasuzzoha, A., Ndzibah, E., & Timilsina, B. (2022). End-to-end servitization model in Industry 4.0. Management and Production Engineering Review, 13(1), 89-98.

Sjödin, D., Kamalaldin, A., Parida, V., & Islam, N. (2023). Procurement 4.0: How industrial customers transform procurement processes to capitalize on digital

servitization. IEEE Transactions on Engineering Management, 70(12), 4175-4190. Sklyar, A., Kowalkowski, C., Tronvoll, B., & Sörhammar, D. (2019). Organizing for digital servitization: A service ecosystem perspective. Journal of Business Research, 104, 450_460

Smania, G. S., de Sousa Mendes, G. H., Godinho Filho, M., Osiro, L., Cauchick-Miguel, P. A., & Coreynen, W. (2022). The relationships between digitalization and ecosystem-related capabilities for service innovation in agricultural machinery manufacturers. Journal of Cleaner Production, 343, Article 130982.

Stentoft, J., & Rajkumar, C. (2020). The relevance of Industry 4.0 and its relationship with moving manufacturing out, back and staying at home. International Journal of Production Research, 58(10), 2953-2973.

Stentoft, J., & Stegmann, O. (2014). Back-shoring manufacturing: Notes on an important but under-researched theme. Journal of Purchasing and Supply management, 20, 60-62.

Stentoft, J., Olhager, J., Heikkilä, J., & Thoms, L. (2016). Manufacturing backshoring: A systematic literature review. Operations Management Research, 10(3), 1-10.

Stentoft, J., Rajkumar, C., & Madsen, E. S. (2017). Industry 4.0 in Danish industry: An empirical investigation of the degree of knowledge, perceived relevance and current practice. Syddansk Universitet.

J. Bilbao-Ubillos et al.

European Research on Management and Business Economics 30 (2024) 100234

- Stentoft, J., Wickstrøm, K. A., Haug, A., & Philipsen, K. (2021). Cost-driven motives to relocate manufacturing abroad among small-and medium-sized manufacturers: The influence of Industry 4.0. *Journal of Manufacturing Technology Management*, 32(3), 646–666.
- Strange, R., & Zucchella, A. (2018). Industry 4.0, global value chains and international business. *Multinational Business Review*, 25(3), 174–184.
- Taleb, T., Afolabi, I., & Bagaa, M. (2022). Orchestrating 5G network slices to support industrial internet and to shape next-generation smart factories. IEEE Network, 33(4), 146–154.
- Tukker, A. (2004). Eight types of product-service system: Eight ways to sustainability? Experiences from suspronet. Business Strategy and the Environment, 13(4), 246–260.
- Unterberger, P., & Müller, J. M. (2021). Clustering and classification of manufacturing enterprises regarding their Industry 4.0 reshoring incentives. *Procedia Computer Science*, 180, 696–705.
- Urzelai, B, & Puig, F (2017). Clústeres como plataforma para la internacionalización. Emprendimiento y negocios internacionales, 2(1), 9–19.
- Vaillant, Y., Lafuente, E., Horváth, K., & Vendrell-Herrero, F. (2021). Regions on course for the fourth industrial revolution: The role of a strong indigenous t-kibs sector. *Regional Studies*, 55(10), 1816–1828.
- Van Hoek, R., & Dobrzykowski, D. (2021). Towards more balanced sourcing strategies are supply chain risks caused by the COVID-19 pandemic driving reshoring considerations? Supply Chain Management, 26(6), 689–701.
- Vandermerwe, S., & Rada, J. (1988). Servitization of business: Adding value by adding services. European Management Journal, 6(4), 314–324.

- Vendrell-Herrero, F., Bustinza, O. F., & Vaillant, Y. (2021). Adoption and optimal configuration of smart products: The role of firm internationalization and offer hybridization. *Industrial Marketing Management*, 95, 41–53.
- Vendrell-Herrero, F., Bustinza, O. F., Parry, G., & Georgantzis, N. (2017). Servitization, digitization and supply chain interdependency. *Industrial Marketing Management*, 60, 69–81.
- Wang, Z., Cui, L., Guo, W., Zhao, L., Yuan, X., Gu, X., Tang, W., Bu, L., & Huang, W. (2022). A design method for an intelligent manufacturing and service system for rehabilitation assistive devices and special groups. *Advanced Engineering Informatics*, 51, Article 101504.
- Weking, J., Stöcker, M., Kowalkiewicz, M., Böhm, M., & Krcmar, H. (2020). Leveraging Industry 4.0–A business model pattern framework. *International Journal of Production Economics*, 225, Article 107588.
- Wen, X., & Zhou, X. (2016). Servitization of manufacturing industries based on cloudbased business model and the down-to-earth implementary path. *The International Journal of Advanced Manufacturing Technology*, 87(5), 1491–1508.
- Wiesmann, B., Snoei, J. R., Hilletofth, P., & Eriksson, D. (2017). Drivers and barriers to reshoring: A literature review on offshoring in reverse. *European Business Review*, 29 (1), 15–42.
- Zheng, P., Wang, Z., Chen, Ch., & Khoo, L. (2019). A survey of smart product-service systems: Key aspects, challenges and future perspectives. *Advanced Engineering Informatics*, 42(C), Article 100973.